Author’s response to reviews

Title: Association between health behaviors and mood disorders among the elderly: a community-based cohort study

Authors:

Tzu-Jung Tseng (jlucy1002@gmail.com)
Yi-Syuan Wu (pu1254@gmail.com)
Jia-Hong Tang (jhtang.cw@gmail.com)
Yen-Hui Chiu (A3360@tpech.gov.tw)
Yu-Ting Lee (rimuy@gate.sinica.edu.tw)
I-Chun Fan (mhfanbbc@ccvax.sinica.edu.tw)
Ta-Chien Chan (dachianpig@gmail.com)

Version: 4 Date: 14 Feb 2019

Author’s response to reviews:

Editor Comments:

1. Please include the email addresses of all authors on the title page of your manuscript.
Response: We have added the e-mail addresses of all authors on the title page.

2. Thank you for including in your Methods section information regarding ethics approval and informed consent. Please also include statement about ethics approval in the Ethics approval section of the Declarations.
Response: We have included the ethics approval in the Ethics approval section of the Declarations.

“This study was approved by the Institutional Review Board (IRB) on Biomedical Science Research, Academia Sinica (AS-IRB02-104182)”

3. Please state in the cover letter whether the maps depicted in Figure 3 are your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.
Response: The figure 3 is drawn by ourselves.
If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time. Should you require an alternative source you may wish to try Wikimedia Commons: http://commons.wikimedia.org/wiki/Main_Page

4. We notice that there are two sets of figures: one set in the main manuscript and one set uploaded to the file inventory. Please remove one of these sets of figures. When doing so, please also ensure that the Figure legends list is still in the main manuscript.

Response: We have removed the figures in the main manuscript and left three separate figures in the system. The figure legend is still in the main manuscript.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: There are only one clean manuscript and three separate figure files.